Author's response to reviews

Title: Evaluation of antinociceptive effect of methanolic extract of leaves of Crataeva nurvala Buch.-Ham.

Authors:

Md. Moniruzzaman (moniruzzaman.babu@yahoo.com)
Mohammad Zafar Imam (zafarimam@gmail.com)

Version: 2 Date: 23 August 2014

Author's response to reviews: see over
To
The Editor
BMC Complementary and Alternative Medicine

Subject: **SUBMISSION OF REVISED MANUSCRIPT FOR EVALUATION**

Dear Sir

I am enclosing herewith a revised manuscript entitled “**Evaluation of antinociceptive effect of methanolic extract of leaves of Crataeva nurvala Buch.-Ham.**” for possible evaluation to publish in “**BMC Complementary and Alternative Medicine**”. We have brought about changes as per the comments of the reviewers. However, As the hot plate and tail immersion test results reflects two slightly different molecular observations we are still including both. We agree with the reviewer but we would like to keep the tables as these give more specific values rather than figures. We also tried to keep the issues of the extract within the discussion of methods. About the references of Substance P of we have mentioned a number of them at the end of the next line and we only mention the referenced that we followed in case of ATP-sensitive K+ channel pathway test.

With thanks and regards

**Mohammad Zafar Imam**
Senior Lecturer
Department of Pharmacy
Stamford University Bangladesh
51, Siddeswari Road, Dhaka-1217
Bangladesh.
E-mail: zafarimam@gmail.com
Tel: +8801816684184